REVIEW

# Therapeutic potential of rivaroxaban in the prevention of venous thromboembolism following hip and knee replacement surgery: a review of clinical trial data

Louis M Kwong

Department of Orthopaedic Surgery, Harbor-UCLA Medical Center, Torrance, CA, USA Abstract: Rivaroxaban (Bayer AG, Leverkusen, Germany) is a highly selective direct inhibitor of factor Xa. It has completed Phase III clinical trials evaluating its efficacy and safety against enoxaparin in the prophylaxis against venous thromboembolism (VTE) in orthopedic patients following primary total hip and total knee arthroplasty. Rivaroxaban has been extensively studied worldwide in 12,729 patients in the Regulation of Coagulation in Major Orthopedic Surgery Reducing the Risk of DVT and PE (RECORD) program. Pivotal clinical trials have demonstrated the superior efficacy in reducing total VTE in comparison with both the North American and European regimens of enoxaparin. Safety of the drug was found to be excellent, with no demonstrable cardiovascular or hepatic effects and no statistically significant increase in major bleeding. A pooled analysis of data collected on the patients from the four RECORD trials revealed rivaroxaban to be the first antithrombotic agent to demonstrate superiority over another antithrombotic (enoxaparin) in reducing symptomatic VTE and all-cause mortality. While there was a significant difference in the composite safety endpoint of major and clinically relevant nonmajor bleeding in the pooled analysis with the use of rivaroxaban compared with enoxaparin, there was no significant difference in major bleeding or in any other bleeding. Keywords: rivaroxaban, venous thromboembolism, hip replacement, knee replacement

## Introduction

Venous thromboembolism (VTE) is the most frequent serious complication following total hip and total knee arthroplasty. In the absence of thromboprophylaxis, 50% to 60% of these patients will develop either deep vein thrombosis (DVT) or pulmonary embolism (PE). Because patients following total joint arthroplasty are at very highest risk for the development of VTE, the use of routine thromboprophylaxis has been recommended in these patients by a number of nationally and internationally recognized entities including the American College of Chest Physicians, the American Academy of Orthopaedic Surgeons, the International Surgical Thrombosis Forum, and the US Centers for Medicare and Medicaid Services.

# **Contemporary antithrombotic strategies**

Current guideline recommendations for thromboprophylaxis involve the use of vitamin K antagonists (VKA), low-molecular-weight heparins (LMWH), aspirin, or fondaparinux, an indirect inhibitor of factor Xa.<sup>1,2,3</sup> While aspirin has been advocated by a number of organizations for thromboprophylaxis,<sup>3</sup> this agent functions principally as an inhibitor of platelet aggregation rather than as an antithrombotic agent. The use

Correspondence: Louis M Kwong MD, FACS 1000 W Carson Street, Box 422 Torrance, CA 90509, USA Tel +1 310 222 2716 Email Imkwongmd@hotmail.com

submit your manuscript | www.dovepress.com Dovepress http://dx.doi.org/10.2147/VHRM.S4441 Vascular Health and Risk Management 2011:7 461–466 © 2011 Kwong, publisher and licensee Dove Medical Press Ltd. This is an Open Access article which permits unrestricted noncommercial use, provided the original work is properly cited. of VKAs is associated with extremely variable patientto-patient response, delay in onset of action, extensive food and drug interactions, and the need for monitoring and dose adjustment to target the international normalized ratio value.<sup>3</sup> Both LMWHs and fondaparinux require subcutaneous injection for drug administration. This can significantly affect patient acceptance and compliance during extended prophylaxis and in circumstances depending on patient self-administration.<sup>4</sup>

#### A new antithrombotic agent

Rivaroxaban is a highly selective direct inhibitor of factor Xa that has completed Phase III clinical trials evaluating its efficacy and safety against enoxaparin in the prophylaxis against VTE in orthopedic patients following primary total hip and total knee arthroplasty. Rivaroxaban has a half-life of approximately 9 hours, with a rapid onset of action, reaching maximum serum concentration in <2 hours. It has very predictable pharmacokinetics and pharmacodynamics, and has no significant food or drug interactions. Approximately one-third of the drug is metabolized by the liver and/or through gastrointestinal transit and eliminated via the fecal route; the remaining two-thirds is cleared via renal excretion.5-7 Rivaroxaban is administered orally at a fixed dose, unmonitored, once daily. As such, the use of the drug provides a very simple regimen for both the patient and the physician, with a route of administration that is readily embraced by the patient, reducing compliance concerns associated with parenteral drug administration.

## **Clinical trial results**

Rivaroxaban has been extensively studied in 12,729 patients worldwide in the RECORD Program (REgulation of Coagulation in major Orthopaedic surgery reducing the Risk of DVT and PE). This represents the largest orthopedic clinical development program ever conducted. Four Phase III pivotal trials evaluated the safety and efficacy of rivaroxaban against the active comparator, enoxaparin. All four clinical trials shared the same basic, prospective, randomized, double-blind study design template, as well as the same blinded, independent adjudication committee for outcomes. These trials also shared the same primary efficacy endpoint, which was a composite of symptomatic DVT, nonfatal PE, and all-cause mortality. All four trials also shared common safety endpoints for major bleeding, major bleeding including the surgical site requiring reoperation, major and clinically relevant nonmajor bleeding,

and any bleeding. This allowed a pooled data analysis from the >12,000 patients of the four pivotal RECORD trials to be performed.

Two of these trials (RECORD1 and 2) dealt with patients following total hip arthroplasty, and two (RECORD3 and 4) dealt with patients following total knee arthroplasty. RECORD1 compared rivaroxaban 10 mg once daily for 5 weeks against enoxaparin 40 mg once daily also for 5 weeks.8 RECORD2 evaluated the safety and efficacy of rivaroxaban 10 mg once daily for 5 weeks against enoxaparin 40 mg once daily for 10-14 days followed by oral placebo.9 In RECORD3, rivaroxaban 10 mg once daily for 10-14 days was compared against enoxaparin 40 mg once daily also for 10-14 days.<sup>10</sup> In RECORD4, rivaroxaban 10 mg once daily for 10-14 days was compared against enoxaparin 30 mg twice daily also for 10-14 days.<sup>11</sup> In all four RECORD trials, rivaroxaban was found to be statistically significantly superior to enoxaparin in the reduction of total VTE (the primary efficacy endpoint). Additionally, in RECORD1, 2, and 3, rivaroxaban was superior in reducing major VTE (a composite of proximal DVT, symptomatic PE, and all-cause mortality), and in RECORD2 and 3, superior in reducing symptomatic VTE (a composite of symptomatic DVT, symptomatic PE, and VTE-related death). RECORD4 is notable in that it represents the only clinical trial ever to demonstrate superior efficacy of any antithrombotic agent in comparison with the North American regimen of enoxaparin (30 mg subcutaneously twice daily). The superior efficacy of rivaroxaban was achieved without a statistically significant increase in major bleeding, which was <0.7% in all trials. Table 1 summarizes the efficacy results from the four RECORD Trials.8-11 Table 2 Summarizes key primary as well as composite safety endpoints. There was no statistically significant difference in major bleeding, or difference in clinically relevant nonmajor bleeding.12

 Table I Summary table RECORD trials: RRR (%) rivaroxaban vs

 enoxaparin

	Total VTE	Major VTE	Symptomatic VTE	Major bleeding*
RECORDI	77	88	NS	NS
RECORD2	79	88	80	NS
RECORD3	49	62	66	NS
RECORD4	31	NS	NS	NS

Note: \*Major bleeding <0.7% in all trials in all groups.

Data from Lassen et al poster.12

**Abbreviations:** NS, no statistically significant difference; VTE, venous thromboembolism.

	Rivaroxaban (n = 6183)	Enoxaparin (n = 6200)	Hazard ratio (95% CI)	P-value
	(n) %	(n) %		
Major bleeding	24 (0.4)	13 (0.2)	1.84 (0.94–3.62)	0.08
Bleeding leading to reoperation	12 (0.2)	7 (0.1)	_	_
Clinically overt extra-surgical site bleeding leading to – transfusion of >2 units of whole blood or PRBC	8 (0.1)	I (<0.1)	-	-
Clinically relevant nonmajor bleeding	177 (2.9)	145 (2.3)	1.22 (0.98-1.52)	0.08
Surgical site bleeding	47 (0.8)	49 (0.8)	-	_
Excessive wound hematoma	53 (0.9)	58 (0.9)	-	-
Major bleeding plus clinically relevant nonmajor bleeding	201 (3.2)	158 (2.5)	_	0.039

#### **Table 2** Safety data at follow-up day $12 \pm 2$ days

Data from Lassen et al poster.<sup>12</sup>

Abbreviation: PRBC, packed red blood cells.

# **Pooled analysis**

A pooled analysis of the data from RECORD1, 2, 3, and 4 was performed to compare rivaroxaban with enoxaparin regimens in reducing the incidence of efficacy and safety-related endpoints across the four pivot trials. A pre-specified integrated analysis strategy for symptomatic event composites across the RECORD1–4 studies was utilized. Because the number of VTE events was expected to be low in any of the individual RECORD studies, pooling was utilized to increase statistical precision, and to allow for a better assessment of clinically relevant endpoints. Because the RECORD trials differed in the duration of drug administration, treatment duration and total follow-up duration, three population pools were evaluated. The Active Treatment Pool represented that group of patients receiving drug administration up to  $12 \pm 2$  days. The Total Treatment Duration Pool included all patients during their treatment periods (from day  $12 \pm 2$  days to day  $35 \pm 4$  days). The total study duration pool represented all patients up to and including the final follow-up (from day 42 to 65). Figure 1 illustrates the pooled analysis study design.<sup>13</sup>

The prespecified primary efficacy outcome in this pooled analysis was a composite of symptomatic DVT and nonfatal



**Figure I** Pooled analysis design. Data from Kwong LM presentation.<sup>13</sup>





PE, as well as all-cause mortality. All symptomatic events were confirmed and centrally adjudicated. Other outcomes evaluated included all-cause mortality; a post-hoc analysis of a composite of PE and all-cause mortality; and a post-hoc analysis of a composite of all major clinical outcomes (death, myocardial infarction, stroke, symptomatic VTE, and major bleeding). Safety outcomes consisted of the prespecified







Figure 3 Active treatment pool efficacy outcome.

**Notes:** Homogeneity test, P = 0.431; safety population, n = 12,383. **Abbreviations:** VTE, venous thromboembolism; RRR, relative risk reduction; HR; hazard ratio.

treatment-emergent bleeding endpoints of major bleeding: major bleeding including surgical site bleeding, major bleeding and clinically relevant nonmajor bleeding, and any bleeding. Other safety endpoints included adverse events, including cardiovascular events.

Rivaroxaban demonstrated a 58% reduction in the primary efficacy outcome (symptomatic VTE and all-cause



Symptomatic VTE and all-cause mortality (during treatment and follow-up)

Figure 4 Total study duration efficacy outcome.

Abbreviations: VTE, venous thromboembolism; HR, hazard ratio.

Table 3 Liver safety: total duration study pool

n/N (%)	Enoxaparin regimens (n = 6200)	Rivaroxaban regimens (n = 6183)		
Abnormalities after day 0 (preop baseline)				
$ALT > 3 \times ULN$	227/6,131 (3.7)	152/6,131 (2.5)		
$ALT > 3 \times ULN + TB > 2 \times ULN$	7/6,131 (0.11)	9/6,131 (0.15)		
Abnormalities after day I (post-op baseline)				
$ALT > 3 \times ULN$	188/6,009 (3.1)	119/6,032 (2.0)		
$ALT > 3 \times ULN + TB > 2 \times ULN$	7/6,009 (0.12)	7/6,032 (0.12)		

Note: All measurements after baseline considered, safety population n=12,383 Data from Lassen et al poster.  $^{12}\,$ 

Abbreviations: ALT; alanine transaminase; ULN, upper limit of normal; TB, total bilirubin.

mortality) for the Total Treatment Duration Pool compared with enoxaparin in the primary population for analysis (Figure 2). In the Active Treatment Pool, there was a 52% reduction in the primary efficacy outcome with rivaroxaban compared with enoxaparin (hazard ratio [HR] 0.48; 95% confidence interval [CI] 0.31-0.75) (Figure 3). Rivaroxaban was also found to significantly reduce the incidence of symptomatic VTE and all-cause mortality in the Total Study Duration Pool up to and including final follow-up (HR 0.49; 95% CI 0.35-0.69) (Figure 4). Additionally, rivaroxaban significantly improved the composite outcome of symptomatic VTE, cardiovascular events, all-cause mortality, and major bleeding compared with enoxaparin regimens. There were no demonstrable adverse effects on hepatic function (Table 3). There was no statistically significant difference in any bleeding, major bleeding, or major plus clinically relevant nonmajor (CRNM) bleeding at day 12  $\pm$  2 days and at follow-up. There was a significant difference in major plus CRNM at day 35 (including the placebo phase of RECORD2) favoring the use of enoxaparin in the safety analysis of composite bleeding endpoints (rivaroxaban 158/12,383 [2.55%]; enoxaparin 197/12,383 [3.19%], P = 0.039 [analyzed using Cox regression model]).12

## Discussion

Rivaroxaban is a new oral antithrombotic agent that is a highly selective direct inhibitor of factor Xa. It has been designed with a simple once-a-day regimen to be used across a broad spectrum of the population at a fixed dose, unmonitored. Pivotal clinical trials have demonstrated the superior efficacy in reducing total VTE in comparison with both the North American and European regimens of enoxaparin. Safety of the drug was found to be excellent with no demonstrable cardiovascular or hepatic effects,

and no statistically significant increase in major bleeding or in any bleeding. Although a statistically significant difference was found in major plus CRNM at day 35 favoring the use of enoxaparin in the safety analysis of composite bleeding endpoints, in the clinical setting, this finding may not represent a relevant factor for the surgeon in terms of surgical site complications (no statistically significant differences were found in bleeding leading to reoperation, prolonged wound drainage, hemarthrosis, or wound dehiscence) and no statistically significant difference was found in postoperative infection rate between the rivaroxaban and enoxaparin groups.<sup>12</sup> The use of pharmacologic thromboprophylaxis as a risk factor for postoperative bleeding and prolonged wound drainage leading to infection is a concern of some orthopedic surgeons, although a causal relationship has never been established, and a recent review of the peer-reviewed literature also does not support this association.14 A pooled analysis of the 12,729 patients from the four RECORD trials revealed rivaroxaban to be the first antithrombotic agent to demonstrate superiority over another antithrombotic (enoxaparin) in reducing symptomatic VTE and all-cause mortality. A pharmacoeconomic analysis of the efficacy and safety of rivaroxaban compared with the use of enoxaparin indicates that rivaroxaban has the potential to be cost effective in reducing costs associated with the prophylaxis and treatment of thromboembolic events in postoperative hip and knee replacement patients.<sup>15,16</sup>

In summary, rivaroxaban appears to have the potential to provide a safe, convenient, and effective method for prophylaxis against VTE in the high-risk group of orthopedic patients following total hip and knee arthroplasty.

#### Disclosures

Dr Kwong has received research grants from Astellas, Bayer, GlaxoSmithKline, Pfizer, sanofi-aventis, and Takeda. He is a consultant for Zimmer and also receives royalties for product development.

#### References

- Geerts WH, Bergqvist D, Pineo GF, et al. Prevention of venous thromboembolism: American College of Chest Physicians Evidence-Based Clinical Practice Guidelines (8th ed). *Chest.* 2008;133:381S–453S.
- Johanson NA, Lachiewicz PF, Leiberman JR, et al; AAOS Physician Volunteer Work Group. American Academy of Orthopaedic Surgeons clinical guideline on prevention of symptomatic pulmonary embolism in patients undergoing total hip or knee arthroplasty: summary of recommendations. J Am Acad Orthop Surg. 2009;17(3):183–196.
- 3. Ellis RF, Stephens MA, Sharp GB. Evaluation of a pharmacy-managed warfarin-monitoring service to coordinate inpatient and outpatient therapy. *Am J Hosp Pharm.* 1992;49(2):387–394.

- Watts AC, Howie CR, Simpson A. Assessment of a self-administration protocol for extended subcutaneous thromboprophylaxis in lower limb arthroplasty. *J Bone Joint Surg.* 2006;88(1):107–110.
- Roehrig S, Straub A, Pohlmann J, et al. Discovery of the novel antithrombotic agent 5-chloro-N-({(5S)-2-oxo-3-[4-(3-oxomorpholin-4-yl)phenyl]-1,3-oxazolidin-5-yl}methyl)thiophene-2-carboxamide (BAY 59-7939): an oral, direct factor Xa inhibitor. *J Med Chem.* 2005;48(19):5900–5908.
- Perzborn E, Kubitza D, Misselwitz F. Rivaroxaban. A novel, oral, direct factor Xa inhibitor in clinical development for the prevention and treatment of thromboembolic disorders. *Hamostaseologie*. 2007;27:282–289.
- Kubitza D, Becka M, Wensing G, Voith B, Zuehlsdorf M. Safety, pharmacodynamics, and pharmacokinetics of BAY 59-7939-an oral, direct factor Xa inhibitor-after multiple dosing in healthy male subjects. *Eur J Clin Pharmacol*. 2005;61(12):873–880.
- Eriksson BI, Borris LC, Friedman RJ, et al. Rivaroxaban versus enoxaparin for thromboprophylaxis after hip arthroplasty. *N Engl J Med.* 2008;358(26):2765–2775.
- Kakkar AK, Brenner B, Dahl OE, et al. Extended duration rivaroxaban versus short-term enoxaparin for the prevention of venous thromboembolism after total hip arthroplasty: a double-blind, randomised controlled trial. *Lancet*. 2008;372(9632):31–39.

- Lassen MR, Ageno W, Borris LC, et al. Rivaroxaban versus enoxaparin for thromboprophylaxis after total knee arthroplasty. *N Engl J Med.* 2008;358(26):2776–2786.
- Turpie AG, Lassen MR, Davidson BL, et al. Rivaroxaban Versus Enoxaparin for Thromboprophylaxis After Total Knee Arthroplasty (RECORD4): a randomised trial. *Lancet*. 2009; 373(9676): 1673–1680.
- Lassen MR, Turpie AG, Kakkar AK, Eriksson BI, Gent M. Potential impact of rivaroxaban on surgical safety outcomes after total hip or knee replacement surgery. Annual Meeting of the International Society on Thrombosis and Haemostasis. PP-WE-404. July 2009.
- Kwong LM. Once-daily, oral rivaroxaban for the prevention of venous thromboembolism (VTE) after total hip or knee replacement (THR/TKR) surgery. Western Orthopaedic Association Annual Meeting. Podium presentation: PM Session. Seattle, WA. August 1, 2009.
- Kwong LM. Bleeding and Thrombo. Poster presentation # 46. 2011 Mid-America Orthopaedic Association Annual Meeting. April 7–8, 2011
- Kwong LM. Cost-effectiveness of rivaroxaban after total hip or knee arthroplasty Am J Manag Care. 2011;17(Suppl):S22–S26.
- Kwong LM, Duran A, Diamantopoulos A, Sengupta N, Lees M. Cost-effectiveness of rivaroxaban for prevention of venous thromboemebolism (VTE) after total hip or knee replacement (THR, TKR) in the US. [Abstract]. *J Thromb Haemost*. 2009;7(Suppl 2):PP-WE-473.

#### Vascular Health and Risk Management

#### Publish your work in this journal

Vascular Health and Risk Management is an international, peerreviewed journal of therapeutics and risk management, focusing on concise rapid reporting of clinical studies on the processes involved in the maintenance of vascular health; the monitoring, prevention and treatment of vascular disease and its sequelae; and the involvement of

the processes involved nitoring, prevention and ; and the involvement of

Submit your manuscript here: http://www.dovepress.com/vascular-health-and-risk-management-journal

metabolic disorders, particularly diabetes. This journal is indexed on

PubMed Central and MedLine. The manuscript management system

## Dovepress